Author’s response to reviews

Title: Overview of the Consortium of Hospitals Advancing Research on Tobacco (CHART)

Authors:

William T Riley (wiriley@mail.nih.gov)
Victor J Stevens (Victor.Stevens@kpchr.org)
Shu-Hong Zhu (szhu@ucsd.edu)
Glen Morgan (gmorgan@mail.nih.gov)
Debra Grossman (dgrossma@mail.nih.gov)

Version: 2 Date: 24 October 2011

Author’s response to reviews:

Trials Editor,

As per your request of October 14, we have revised the manuscript to conform to journal style. In the revised manuscript, we have included competing interest, acknowledgement, and author’s contribution sections. We also note here that this overview article does not have a trials registration or human subjects approval documentation because this article is an overview of the individual project manuscripts, each of which has its own trials registration number and human subjects approval documentation.